```
Mauna Loa Solar Observatory Observer's Log
```

\_\_\_\_\_ Wed Oct 1 16:31:15 GMT 2003 Year: 03 Doy: 274 Observer: yasukawa WEATHER COMMENT: Wed Oct 1 16:31:50 GMT 2003 Cool, in moderately thick cirrostratus, south wind. Wed Oct 1 16:47:49 GMT 2003 CHIP Start Patrol Wed Oct 1 16:48:02 GMT 2003 PTCS Start Patrol PSPT COMMENT: Wed Oct 1 17:06:25 GMT 2003 Starting up. \*\*MKIV PROBLEM\*\*: Wed Oct 1 17:06:49 GMT 2003 Nahenahe keyboard not communicating with workstation at this time. investigating. PSPT COMMENT: Wed Oct 1 17:10:14 GMT 2003 observing \_\_\_\_\_ Mauna Loa Solar Observatory Observer's Log \_\_\_\_\_ Wed Oct 1 17:37:55 GMT 2003 Year: 03 Doy: 274 Observer: yasukawa \*\*MKIV PROBLEM\*\*: Wed Oct 1 17:38:13 GMT 2003 Rebooted nahenahe to recover from keyboard loss. Wed Oct 1 18:02:35 GMT 2003 PICS Flat Wed Oct 1 18:02:55 GMT 2003 CHIP LSD Wed Oct 1 18:04:39 GMT 2003 CHIP End LSD Wed Oct 1 18:04:45 GMT 2003 CHIP BiasLSD Wed Oct 1 18:04:56 GMT 2003 End Flat PICS Wed Oct 1 18:05:36 GMT 2003 CHIP End BiasLSD Wed Oct 1 18:05:43 GMT 2003 CHIP Bias Wed Oct 1 18:06:34 GMT 2003 CHIP End Bias \*\*\*\* EVENT COMMENT \*\*\*\*: Wed Oct 1 20:57:30 GMT 2003 Active region between 1930-2030 UT near CMP, in southern hemisphere. Wed Oct 1 21:30:59 GMT 2003 CHIP End Patrol COMMENT: Wed Oct 1 21:31:11 GMT 2003 Thought I'd run with the dome flap connected to the shutter to see if we are getting close to the season where we don't have to reconfigure the slot. CHIP got domed out. A bit too early to discontinue reconfiguring dome shutter. Wed Oct 1 21:33:50 GMT 2003 PICS End Patrol Wed Oct 1 21:42:42 GMT 2003 CHIP Start Patrol Wed Oct 1 21:42:52 GMT 2003 PICS Start Patrol COMMENT: Wed Oct 1 21:46:26 GMT 2003 Reconfigured dome shutter. COMMENT: Wed Oct 1 23:24:08 GMT 2003 Ran into ladder again. Thu Oct 2 00:32:32 GMT 2003 CHIP End Patrol

COMMENT: Thu Oct 2 00:32:38 GMT 2003 Stopping to reconfigure dome shutter. Thu Oct 2 00:35:00 GMT 2003 PICS End Patrol Thu Oct 2 00:42:31 GMT 2003 CHIP Start Patrol Thu Oct 2 00:42:41 GMT 2003 PICS Start Patrol COMMENT: Thu Oct 2 01:50:39 GMT 2003 Declination reset. Repointed and reset guider. PSPT COMMENT: Thu Oct 2 02:23:04 GMT 2003 Shutting down. Thu Oct 2 02:32:17 GMT 2003 CHIP End Patrol Thu Oct 2 02:32:55 GMT 2003 PICS End Patrol \*\*MLSO PROBLEM\*\*: Thu Oct 2 02:32:44 GMT 2003 connection to hp laserprinter (pai) still not re-established. I tried looking for the old IP to change, but was not successful. It would be helpful to get someone versed in these things and knows where to look to help out. Thu Oct 2 02:40:48 GMT 2003 MkIV